Author’s response to reviews

Title: Impact of Pharmacy-Led Medication Reconciliation on Admission to Internal Medicine Service: Experience in Two Tertiary Care Teaching Hospitals

Authors:

Lamis Karaoui (lamis.karaoui@lau.edu.lb)
Nibal Chamoun (nibal.chamoun@lau.edu.lb)
Jessica Fakhir (jjfakhir@stgeorgehospital.org)
Wael Abi Ghanem (weabighanem@stgeorgehospital.org)
Sara El Droubi (sarandroubi@gmail.com)
Abdul Rahman Diab Marzouk (diab.abed@gmail.com)
Nabila Droubi (nabila.droubi@gmail.com)
Hiba Masri (pharm.hiba@gmail.com)
Elsy Ramia (elsy.ramia@lau.edu.lb)

Version: 5 Date: 21 Jun 2019

Author’s response to reviews:

BHSRD-19-00003R4
Impact of Pharmacy-Led Medication Reconciliation on Admission to Internal Medicine Service: Experience in Two Tertiary Care Teaching Hospitals Lamis Karaoui, PharmD; Nibal Chamoun, PharmD; Jessica Fakhir, PharmD; Wael Abi Ghanem, PharmD; Sara El Droubi, PharmD; Abdul Rahman Diab Marzouk, BS Pharm; Nabila Droubi, PharmD; Hiba Masri, PharmD; Elsy Ramia, Pharm.D., MPH BMC Health Services Research

Response to the Editor Comments:

1. Authors’ Contributions
   We note that author Lamis Karaoui is listed in your Authors’ Contributions section as “LR” when it should in fact be “LK”. Please amend this.
   Thank you for your comment.
   The author Lamis Karaoui is now listed as LK in the Author’s Contributions section.

2. Additional files list
   Please remove your cover letter from your additional files list as this will not be published with your manuscript.
   Thank you for your comment.
The corresponding author removed the cover letter from the additional files list.

3. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Thank you for your comment.
The manuscript is now uploaded as a single, final, clean version.
All tables and supplementary files are cited within the text.